“Early onset lung cancer, cigarette smoking and the genetic polymorphism SNP309 of the murine double minute-2 (MDM2) gene” (BMC-Cancer/2007/1996041118160143)

Dear BioMed Central Editorial Production Team,

Thank you for publishing our manuscript in your journal.

We have itemized our minor changes below in a point by point fashion and have revised the manuscript accordingly. All authors have read the revised paper.

Thank you very much for your efforts.

Sincerely Yours,

Kirstin Mittelstraß

# Description of changes:

**Spelling:** We have corrected some minor spelling mistakes and also changed the word “sex” with the word “gender” in the whole document.

**Tables:** The back slash between the different genotypes got changed to “vs.”. We also noticed that we had different number of signs after decimal point for our p-values and ORs. Therefore we changed this to three signs after decimal point (p-values) and one sign after decimal point (OR) in the whole document.